Author’s response to reviews

Title: Exploring Multi-level System Factors Facilitating Educator Training and Implementation of Evidence-Based Practices (EBP): A Study Protocol

Authors:

Aubyn Stahmer (astahmer@ucdavis.edu)
Jessica Suhrheinrich (jsuhrheinrich@sdsu.edu)
Patricia Schetter (plschetter@ucdavis.edu)
Elizabeth McGhee Hassrick (emh347@drexel.edu)

Version: 1 Date: 28 Nov 2017

Author’s response to reviews:

Thank you for your initial review of the manuscript "Exploring Multi-level System Factors Facilitating Educator Training and Implementation of Evidence-Based Practices (EBP): A Study Protocol" (IMPS-D-17-00731).

1. The table has been moved to the end of the document.

2. Figure legends, including the title and detailed legend, are now included at the end of the manuscript.

Thank you for your review of the project.

Best,

Aubyn Stahmer